Author's response to reviews

Title: Knowledge, experience, and potential risks of dating violence among Japanese university students: a cross-sectional study

Authors:

Mayumi Ohnishi (mohnishi@nagasaki-u.ac.jp)
Rieko Nakao (rieko@nagasaki-u.ac.jp)
Tomoko Shibayama (lindaom@aol.com)
Yumi Matsuyama (adoresizumu@yahoo.co.jp)
Kazuyo Oishi (k-oishi@nagasaki-u.ac.jp)
Harumi Miyahara (spring@nagasaki-u.ac.jp)

Version: 9 Date: 13 May 2011

Author's response to reviews:

Thank you for your comments. We Japanese usually don't have middle name and use only one family mane. Therefore we don’t have an initial. For example, Mayumi is my first name and Ohnishi is my family name.

We carefully checked the latest version of manuscript.